Author's response to reviews

Title: Suspected metastatic Adrenocortical Carcinoma revealing as Pulmonary Kaposi Sarcoma in Adrenal Cushing`s syndrome

Authors:

Margarita Bala (bala_m@ukw.de)
Cristina L Ronchi (ronchi_c@ukw.de)
Josef Pichl (josef.pichl@theresien-krankenhaus.de)
Vanessa Wild (E_Wild_V@ukw.de)
Stefan Kircher (kircher_s@ukw.de)
Bruno Allolio (allolio_b@ukw.de)
Stefanie Hahner (hahner_s@ukw.de)

Version: 4
Date: 21 July 2014

Author's response to reviews:

Dear Ms. Cruz,

Please find attached the revised version of our manuscript (editor’s request):

Suspected metastatic Adrenocortical Carcinoma revealing as Pulmonary Kaposi Sarcoma in Adrenal Cushing`s syndrome.

We separated the images into 5 figures and added legends for 5 figures in the text accordingly.

Sincerely,

Margarita Bala, MD